Author's response to reviews

Title: MTDH Mediates Trastuzumab Resistance in HER2 Positive Breast Cancer by Decreasing PTEN Expression through an NF-kappa B dependent Pathway.

Authors:

cheng du (dc1115010@sina.com)
min x yi (yiserpent163@163.com)
chao w liu (xicancer@fmmu.edu.cn)
tao han (doctor_than@163.com)
zhe z liu (doctor_lzz@163.com)
yu z ding (doctor_dzy@163.com)
dong z zheng (doctor_zzdong@163.com)
ying piao (doctor_py@163.com)
lin j yuan (yuanjianlin317@126.com)
dong x xie (doctor_xxd@163.com)
ling y han (hylcardiology@163.com)
jiang m xie (xiemanjiang@163.com)

Version: 2  Date: 23 June 2014

Author's response to reviews:

1. I clarified the ethical approval for animal studies in the methods section of the manuscript.
2. I corrected the names in the submission system to get them match that on the title page of the manuscript.
3. I submitted a revised manuscript in an editable format (word document).